Author’s response to reviews

Title: Building spatial composite indicators to analyse environmental health inequalities on a regional scale

Authors:

SAIB S Mahdi-Salim M.S (s_m1285@hotmail.com)
CAUDEVILLE C Julien J (Julien.CAUDEVILLE@ineris.fr)
CARR C Florence F (Florence.CARRE@ineris.fr)
Beauchamp B Maxime M (Maxime.Beauchamp@ineris.com)
CICOLELLA C Andre A (Andre.CICOLELLA@ineris.fr)
Ganry G olivie O (Ganry.olivier@chu-amiens.fr)
Trugeot T Alain A (Alain.trugeot@or2s.fr)

Version: 2
Date: 25 December 2014

Author’s response to reviews:

Dear Editor and reviewers

Please find enclosed the reviewed manuscript entitled: Building spatial composite indicators to analyse environmental health inequalities on a regional scale.

The manuscript was edited in proper English language, grammar, punctuation, spelling, and overall style highly qualified native English speakers, editors at American Journal Experts.
This paper presents the methods to build the composite indicators to analyse environmental health inequalities.

I hereby certify that this is an original paper, which has not been published elsewhere, that it is not under consideration for publication elsewhere, that its publication is approved by all authors and tacitly or explicitly by the responsible authorities where the work was carried out. All authors have participated in the study and concur with the submission and subsequent revisions submitted by the corresponding author.

You may communicate with me in all aspects of the publication.
Thank you for your cooperation in anticipation.

SAIB-Mahdi-Salim